Author’s response to reviews

Title: What is the role of Ewingella americana in humans? A case report in a healthy 4-year-old girl

Authors:

Susanna Esposito (Susanna.Esposito@unimi.it)
Francesco Miconi (francesco.miconi90@gmail.com)
Daniela Molinari (d.molinari@aospterni.it)
Emanuela Savarese (manu-s84@hotmail.it)
Federica Celi (f.celi@aospterni.it)
Livio Marchese (liviomarchese@gmail.com)
Sara Valloscuro (s.valloscuro@gmail.com)
Giovanni Miconi (g.miconi@aospterni.it)
Nicola Principi (nicola.principi@unimi.it)

Version: 2 Date: 23 Apr 2019

Author’s response to reviews:

Editor

BMC Infectious Diseases

Perugia, April 23, 2019

Dear Editor,
Thank you for appreciating our manuscript # INFD-D-19-00037R1 entitled “What is the role of Ewingella americana in humans? A case report in a healthy 4-year-old girl” which we submitted for publication as Case Report in BMC Infectious Diseases.

Please find attached the revised text and our replies to Editor’s comments and recommendations. The authors, all of whom contributed significantly to the manuscript and declare that they have no potential conflict of interest, have seen and approved the final version of the manuscript.

We declare that the text has been reviewed by a native English speaker with appropriate knowledge of the subject matter. We hope that you will now find the paper suitable for publication in BMC Infectious Diseases.

Yours faithfully,

Susanna Esposito,
Pediatric Clinic,
Department of Surgical and Biomedical Sciences
Università degli Studi di Perugia,
Piazza Menghini 1, 06129 Perugia, Italy
Phone: +39 075 5784417; fax: +39 075 5784415
E-mail: susanna.esposito@unimi.it

--As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

Re: Done as recommended (p. 9).
-- Please ensure that Discussion and Conclusion sections are combined to one section 'Discussion and Conclusion'.

Re: Done as suggested (p. 6).

-- Thank you for providing a CARE document. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

Re: Removed.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Re: Done.